Author’s response to reviews

Title: Fenoldopam use in a burn intensive care unit: a retrospective study

Authors:

john w simmons (simmons.john.w@gmail.com)
kevin k chung (Kevin.chung@amedd.army.mil)
evan m renz (Evan.Renz@amedd.army.mil)
christopher e white (christopher.white@amedd.army.mil)
casey l cotant (Casey.Cotant@lackland.af.mil)
molly a tilley (molly.tilley@us.af.mil)
mark o hardin (mark.hardin@amedd.army.mil)
john a jones (John.jones@amedd.army.mil)
lorne h blackbourne (lorne.blackbourne@amedd.army.mil)
steven e wolf (Steven.Wolf@amedd.army.mil)

Version: 3 Date: 20 June 2010

Author’s response to reviews: see over
To the BMC Editorial Production Team,

We thank you for accepting our manuscript, “Fenoldopam use in a burn intensive care unit: a retrospective study” for publication in *BMC Anesthesiology*. We have made the following formatting changes to the manuscript as requested:

1. In the section authors contributions we have inserted “All authors read and approved the final manuscript.
2. Tables – all tables have been reformatted with vertical lines removed and have been pasted into the manuscript
3. All authors email addresses preceded by the authors initials have been listed separately on the title page
4. The disclaimer has been moved to the end of the manuscript in the acknowledgements section
5. The heading “Introduction” has been changed to “Background” in the abstract section
6. All highlighting has been removed from the manuscript
7. The abbreviations have been reformatted as a paragraph rather than a list
8. The legends and titles for figures have been added to the manuscript following the references

Thank you and please inform us of any further revisions, if necessary.

Sincerely,

John W. Simmons, MD, CPT, USA
United States Army Institute of Surgical Research
3400 Rawley E. Chambers Ave
Fort Sam Houston, TX 78234
Phone: 210-916-3301
Fax: 210-271-0830
john.simmons@amedd.army.mil